Author’s response to reviews

Title: Severe intoxication caused by sodium-glucose cotransporter 2 inhibitor overdose: a case report

Authors:
Miho Nakamura (nakamura-mh@med.kanazawa-u.ac.jp)
Junya Nakade (jnakade-knz@staff.kanazawa-u.ac.jp)
Tadashi Toyama (t-toyama@staff.kanazawa-u.ac.jp)
Masaki Okajima (mmokaji@gmail.com)
Takumi Taniguchi (ttaniyan@med.kanazawa-u.ac.jp)

Version: 2 Date: 24 Dec 2019

Author’s response to reviews:

24 December 2019

Dr. Jing Zhang
Editor
BMC Pharmacology and Toxicology

Subject: Submission of revised paper PHAT-D-19-00267R1, titled “Severe intoxication caused by sodium-glucose cotransporter 2 inhibitor overdose: a case report.”

Dear Dr. Jing Zhang,

Thank you very much for editor’s comments. We have carefully reviewed the comments and have revised the manuscript accordingly.

Our responses are given in a point-by-point manner below.
With these changes to our final manuscript, we hereby resubmit our manuscript for an evaluation. Thank you once again for your consideration of our paper.

Sincerely,

Miho Nakamura, MD

Intensive Care Unit, Kanazawa University Hospital
13-1 Takara-machi, Kanazawa, Ishikawa, 920-8641, Japan
Tel: 81-76-265-2961
Fax: 81-76-234-4339
E-mail: nakamura-mh@med.kanazawa-u.ac.jp

Response to Editor:

Q1. Under your Ethics approval and consent to participate heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

A1. We agree that an approval of the Ethics Committee and informed consent from the patient is important. As we wrote in the previous manuscript, we obtained written informed consent from the patients. Regarding a case report and case-series studies, approval of the Ethics Committee is not required in our facility.

The text has been changed as follows.

Ethics approval and consent to participate: In Kanazawa University Hospital, ethics approval is not required for a case report with written informed consent. (Page 7, line 128-129)
Q2. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

A2. It has been changed as pointed out.

Thank you once again for your valuable suggestions.